Author’s response to reviews

Title: Clinical outcomes associated with catecholamine use in patients diagnosed with Takotsubo Cardiomyopathy

Authors:
Uzair Ansari (uzair.ansari@yahoo.com)
Ibrahim El-Battrawy (ibrahim.el-battrawy@medma.uni-heidelberg.de)
Christian Fastner (christian.fastner@umm.de)
Katherine Sattler (katherine.sattler@umm.de)
Michael Behnes (michael.behnes@umm.de)
Erol Tueluemen (erol.tueluemen@umm.de)
Martin Borggrefe (martin.borggrefe@umm.de)
Ibrahim Akin (ibrahim.akin@umm.de)

Version: 3 Date: 13 Dec 2017

Author’s response to reviews:

Response to reviewer:

Editor Comments:

The revised manuscript has been significantly improved. However, the authors should address only a issue; since table 5 is difficult to read, they should classify or summarize the death causes in the short- and long-term.

Answer: We have added a table classifying the causes of death in the short- and long-term. This was possible by assessment of charts and/or telephonic reviews. We accessed medical records
from treating GP’s or relatives, and if further information was not available addressing the cause of death, this was defined as death due to an unknown cause.